Author's response to reviews

Title: Nursing and Midwifery Regulatory Reform in East, Central, and Southern Africa: A Survey of Key Stakeholders.

Authors:

Carey F McCarthy (vej6@cdc.gov)
Joachim Voss (vossj@uw.edu)
Marla E Salmon (msalmon@uw.edu)
Jessica M Gross (jessicamadigangross@gmail.com)
Maureen A Kelley (vef5@cdc.gov)
Patricia L Riley (pyr0@cdc.gov)

Version: 2 Date: 29 May 2013

Author's response to reviews:

The authors would like to thank the reviewers for their time and careful consideration of the manuscript. All comments have been addressed; details are found in the uploaded document.